\begin{thebibliography}{}
\expandafter\ifx\csname natexlab\endcsname\relax\def\natexlab#1{#1}\fi
\providecommand{\url}[1]{\href{#1}{#1}}

\bibitem[{{Astropy Collaboration} {et~al.}(2018){Astropy Collaboration},
  {Price-Whelan}, {Sip{\H{o}}cz}, {G{\"u}nther}, {Lim}, {Crawford}, {Conseil},
  {Shupe}, {Craig}, {Dencheva}, {Ginsburg}, {Vand erPlas}, {Bradley},
  {P{\'e}rez-Su{\'a}rez}, {de Val-Borro}, {Aldcroft}, {Cruz}, {Robitaille},
  {Tollerud}, {Ardelean}, {Babej}, {Bach}, {Bachetti}, {Bakanov}, {Bamford},
  {Barentsen}, {Barmby}, {Baumbach}, {Berry}, {Biscani}, {Boquien}, {Bostroem},
  {Bouma}, {Brammer}, {Bray}, {Breytenbach}, {Buddelmeijer}, {Burke},
  {Calderone}, {Cano Rodr{\'\i}guez}, {Cara}, {Cardoso}, {Cheedella}, {Copin},
  {Corrales}, {Crichton}, {D'Avella}, {Deil}, {Depagne}, {Dietrich}, {Donath},
  {Droettboom}, {Earl}, {Erben}, {Fabbro}, {Ferreira}, {Finethy}, {Fox},
  {Garrison}, {Gibbons}, {Goldstein}, {Gommers}, {Greco}, {Greenfield},
  {Groener}, {Grollier}, {Hagen}, {Hirst}, {Homeier}, {Horton}, {Hosseinzadeh},
  {Hu}, {Hunkeler}, {Ivezi{\'c}}, {Jain}, {Jenness}, {Kanarek}, {Kendrew},
  {Kern}, {Kerzendorf}, {Khvalko}, {King}, {Kirkby}, {Kulkarni}, {Kumar},
  {Lee}, {Lenz}, {Littlefair}, {Ma}, {Macleod}, {Mastropietro}, {McCully},
  {Montagnac}, {Morris}, {Mueller}, {Mumford}, {Muna}, {Murphy}, {Nelson},
  {Nguyen}, {Ninan}, {N{\"o}the}, {Ogaz}, {Oh}, {Parejko}, {Parley}, {Pascual},
  {Patil}, {Patil}, {Plunkett}, {Prochaska}, {Rastogi}, {Reddy Janga},
  {Sabater}, {Sakurikar}, {Seifert}, {Sherbert}, {Sherwood-Taylor}, {Shih},
  {Sick}, {Silbiger}, {Singanamalla}, {Singer}, {Sladen}, {Sooley},
  {Sornarajah}, {Streicher}, {Teuben}, {Thomas}, {Tremblay}, {Turner},
  {Terr{\'o}n}, {van Kerkwijk}, {de la Vega}, {Watkins}, {Weaver}, {Whitmore},
  {Woillez}, {Zabalza}, \& {Astropy Contributors}}]{Ast18}
{Astropy Collaboration}, {Price-Whelan}, A.~M., {Sip{\H{o}}cz}, B.~M., {et~al.}
  2018, \aj, 156, 123

\bibitem[{{Astudillo-Defru} {et~al.}(2017){Astudillo-Defru}, {Delfosse},
  {Bonfils}, {Forveille}, {Lovis}, \& {Rameau}}]{Def17}
{Astudillo-Defru}, N., {Delfosse}, X., {Bonfils}, X., {et~al.} 2017, \aap, 600,
  A13

\bibitem[{Barnes {et~al.}(2005)Barnes, Cameron, Donati, James, Marsden, \&
  Petit}]{barnes2005dependence}
Barnes, J., Cameron, A.~C., Donati, J.-F., {et~al.} 2005, Monthly Notices of
  the Royal Astronomical Society: Letters, 357, L1

\bibitem[{Barnes {et~al.}(2004)Barnes, James, \&
  Cameron}]{barnes2004differential}
Barnes, J., James, D., \& Cameron, A.~C. 2004, Monthly Notices of the Royal
  Astronomical Society, 352, 589

\bibitem[{{Baroch} {et~al.}(2020){Baroch}, {Morales}, {Ribas}, {Herrero},
  {Rosich}, {Perger}, {Anglada-Escud{\'e}}, {Reiners}, {Caballero},
  {Quirrenbach}, {Amado}, {Jeffers}, {Cifuentes}, {Passegger}, {Schweitzer},
  {Lafarga}, {Bauer}, {B{\'e}jar}, {Colom{\'e}}, {Cort{\'e}s-Contreras},
  {Dreizler}, {Galad{\'\i}-Enr{\'\i}quez}, {Hatzes}, {Henning}, {Kaminski},
  {K{\"u}rster}, {Montes}, {Rodr{\'\i}guez-L{\'o}pez}, \&
  {Zechmeister}}]{Baroch2020}
{Baroch}, D., {Morales}, J.~C., {Ribas}, I., {et~al.} 2020, \aap, 641, A69

\bibitem[{{Bernstein} {et~al.}(2003){Bernstein}, {Shectman}, {Gunnels},
  {Mochnacki}, \& {Athey}}]{Ber03}
{Bernstein}, R., {Shectman}, S.~A., {Gunnels}, S.~M., {Mochnacki}, S., \&
  {Athey}, A.~E. 2003, in Society of Photo-Optical Instrumentation Engineers
  (SPIE) Conference Series, Vol. 4841, \procspie, ed. M.~{Iye} \& A.~F.~M.
  {Moorwood}, 1694--1704

\bibitem[{{Berta} {et~al.}(2012){Berta}, {Irwin}, {Charbonneau}, {Burke}, \&
  {Falco}}]{Ber12}
{Berta}, Z.~K., {Irwin}, J., {Charbonneau}, D., {Burke}, C.~J., \& {Falco},
  E.~E. 2012, \aj, 144, 145

\bibitem[{{Browning} {et~al.}(2010){Browning}, {Basri}, {Marcy}, {West}, \&
  {Zhang}}]{Browning2010}
{Browning}, M.~K., {Basri}, G., {Marcy}, G.~W., {West}, A.~A., \& {Zhang}, J.
  2010, \aj, 139, 504

\bibitem[{Cameron {et~al.}(2017)Cameron, Dikpati, \& Brandenburg}]{Cam17}
Cameron, R.~H., Dikpati, M., \& Brandenburg, A. 2017, Space Science Reviews,
  210, 367.
\newblock \url{https://doi.org/10.1007/s11214-015-0230-3}

\bibitem[{{Catalano} \& {Marilli}(1983)}]{Catalano1983}
{Catalano}, S., \& {Marilli}, E. 1983, \aap, 121, 190

\bibitem[{{Chabrier} \& {Baraffe}(1997)}]{Cha97}
{Chabrier}, G., \& {Baraffe}, I. 1997, \aap, 327, 1039

\bibitem[{{Charbonneau}(2014)}]{Cha14}
{Charbonneau}, P. 2014, \araa, 52, 251

\bibitem[{{Delfosse} {et~al.}(1998){Delfosse}, {Forveille}, {Perrier}, \&
  {Mayor}}]{Del98}
{Delfosse}, X., {Forveille}, T., {Perrier}, C., \& {Mayor}, M. 1998, \aap, 331,
  581

\bibitem[{{Dikpati} \& {Charbonneau}(1999)}]{Dik99}
{Dikpati}, M., \& {Charbonneau}, P. 1999, \apj, 518, 508

\bibitem[{Earl {et~al.}(2021)Earl, Tollerud, Jones, Kerzendorf, O'Steen, Busko,
  shaileshahuja, D'Avella, Robitaille, Ginsburg, Homeier, Sipőcz, Averbukh,
  Ogaz, Geda, Tocknell, Davies, Günther, Cherinka, Barbary, Foster,
  Droettboom, Torres, Bray, Casey, Teuben, Crawford, Ferguson, Lim, \&
  Cruz}]{Nic21}
Earl, N., Tollerud, E., Jones, C., {et~al.} 2021, astropy/specutils: v1.3,
  vv1.3,  Zenodo, doi:10.5281/zenodo.4987675.
\newblock \url{https://doi.org/10.5281/zenodo.4987675}

\bibitem[{{Gastine} {et~al.}(2014){Gastine}, {Yadav}, {Morin}, {Reiners}, \&
  {Wicht}}]{Yad13}
{Gastine}, T., {Yadav}, R.~K., {Morin}, J., {Reiners}, A., \& {Wicht}, J. 2014,
  \mnras, 438, L76

\bibitem[{Hall {et~al.}(2007)Hall, Lockwood, \& Skiff}]{Hal07}
Hall, J.~C., Lockwood, G.~W., \& Skiff, B.~A. 2007, The Astronomical Journal,
  133, 862.
\newblock \url{https://doi.org/10.1086%2F510356}

\bibitem[{{Hawley} {et~al.}(1996){Hawley}, {Gizis}, \& {Reid}}]{Hawley96}
{Hawley}, S.~L., {Gizis}, J.~E., \& {Reid}, I.~N. 1996, \aj, 112, 2799

\bibitem[{{Hawley} \& {Pettersen}(1991)}]{Haw91}
{Hawley}, S.~L., \& {Pettersen}, B.~R. 1991, \apj, 378, 725

\bibitem[{{Houdebine} {et~al.}(2017){Houdebine}, {Mullan}, {Bercu}, {Paletou},
  \& {Gebran}}]{Hou17}
{Houdebine}, E.~R., {Mullan}, D.~J., {Bercu}, B., {Paletou}, F., \& {Gebran},
  M. 2017, \apj, 837, 96

\bibitem[{{Irwin} {et~al.}(2011){Irwin}, {Berta}, {Burke}, {Charbonneau},
  {Nutzman}, {West}, \& {Falco}}]{Irw11}
{Irwin}, J., {Berta}, Z.~K., {Burke}, C.~J., {et~al.} 2011, \apj, 727, 56

\bibitem[{{Kelson}(2003)}]{Kel03}
{Kelson}, D.~D. 2003, \pasp, 115, 688

\bibitem[{{Kelson} {et~al.}(2000){Kelson}, {Illingworth}, {van Dokkum}, \&
  {Franx}}]{Kel00}
{Kelson}, D.~D., {Illingworth}, G.~D., {van Dokkum}, P.~G., \& {Franx}, M.
  2000, \apj, 531, 159

\bibitem[{{Kesseli} {et~al.}(2018){Kesseli}, {Muirhead}, {Mann}, \&
  {Mace}}]{Kesseli2018}
{Kesseli}, A.~Y., {Muirhead}, P.~S., {Mann}, A.~W., \& {Mace}, G. 2018, \aj,
  155, 225

\bibitem[{{Lehtinen} {et~al.}(2020){Lehtinen}, {K{\"a}pyl{\"a}}, {Olspert}, \&
  {Spada}}]{Leh20}
{Lehtinen}, J.~J., {K{\"a}pyl{\"a}}, M.~J., {Olspert}, N., \& {Spada}, F. 2020,
  arXiv e-prints, arXiv:2007.00040

\bibitem[{Lovis {et~al.}(2011)Lovis, S{\'e}gransan, Mayor, Udry, Benz, Bertaux,
  Bouchy, Correia, Laskar, Curto, {et~al.}}]{Lov11}
Lovis, C., S{\'e}gransan, D., Mayor, M., {et~al.} 2011, Astronomy \&
  Astrophysics, 528, A112

\bibitem[{{Magaudda} {et~al.}(2020){Magaudda}, {Stelzer}, {Covey}, {Raetz},
  {Matt}, \& {Scholz}}]{Magaudda2020}
{Magaudda}, E., {Stelzer}, B., {Covey}, K.~R., {et~al.} 2020, \aap, 638, A20

\bibitem[{{Mamajek} \& {Hillenbrand}(2008)}]{Mamajek2008}
{Mamajek}, E.~E., \& {Hillenbrand}, L.~A. 2008, \apj, 687, 1264

\bibitem[{{Mann} {et~al.}(2016){Mann}, {Gaidos}, {Mace}, {Johnson}, {Bowler},
  {LaCourse}, {Jacobs}, {Vanderburg}, {Kraus}, {Kaplan}, \& {Jaffe}}]{Man16}
{Mann}, A.~W., {Gaidos}, E., {Mace}, G.~N., {et~al.} 2016, \apj, 818, 46

\bibitem[{{Medina} {et~al.}(2020){Medina}, {Winters}, {Irwin}, \&
  {Charbonneau}}]{Med20}
{Medina}, A.~A., {Winters}, J.~G., {Irwin}, J.~M., \& {Charbonneau}, D. 2020,
  \apj, 905, 107

\bibitem[{{Middelkoop}(1982)}]{Middelkoop1982}
{Middelkoop}, F. 1982, \aap, 107, 31

\bibitem[{Newton {et~al.}(2017)Newton, Irwin, Charbonneau, Berlind, Calkins, \&
  Mink}]{Newton2017}
Newton, E.~R., Irwin, J., Charbonneau, D., {et~al.} 2017, The Astrophysical
  Journal, 834, 85

\bibitem[{{Newton} {et~al.}(2016){Newton}, {Irwin}, {Charbonneau},
  {Berta-Thompson}, {Dittmann}, \& {West}}]{Newton2016}
{Newton}, E.~R., {Irwin}, J., {Charbonneau}, D., {et~al.} 2016, \apj, 821, 93

\bibitem[{{Newton} {et~al.}(2018){Newton}, {Mondrik}, {Irwin}, {Winters}, \&
  {Charbonneau}}]{Newton2018}
{Newton}, E.~R., {Mondrik}, N., {Irwin}, J., {Winters}, J.~G., \&
  {Charbonneau}, D. 2018, \aj, 156, 217

\bibitem[{{Noyes} {et~al.}(1984){Noyes}, {Hartmann}, {Baliunas}, {Duncan}, \&
  {Vaughan}}]{Noy84}
{Noyes}, R.~W., {Hartmann}, L.~W., {Baliunas}, S.~L., {Duncan}, D.~K., \&
  {Vaughan}, A.~H. 1984, \apj, 279, 763

\bibitem[{{N{\'u}{\~n}ez} {et~al.}(2015){N{\'u}{\~n}ez}, {Ag{\"u}eros},
  {Covey}, {Hartman}, {Kraus}, {Bowsher}, {Douglas}, {L{\'o}pez-Morales},
  {Pooley}, {Posselt}, {Saar}, \& {West}}]{Nunez2015}
{N{\'u}{\~n}ez}, A., {Ag{\"u}eros}, M.~A., {Covey}, K.~R., {et~al.} 2015, \apj,
  809, 161

\bibitem[{{Nutzman} \& {Charbonneau}(2008)}]{Nut08}
{Nutzman}, P., \& {Charbonneau}, D. 2008, \pasp, 120, 317

\bibitem[{{Owen} \& {Mohanty}(2016)}]{Owe16}
{Owen}, J.~E., \& {Mohanty}, S. 2016, \mnras, 459, 4088

\bibitem[{{Pallavicini} {et~al.}(1981){Pallavicini}, {Golub}, {Rosner},
  {Vaiana}, {Ayres}, \& {Linsky}}]{Pal81}
{Pallavicini}, R., {Golub}, L., {Rosner}, R., {et~al.} 1981, \apj, 248, 279

\bibitem[{{Parker}(1955)}]{Par55}
{Parker}, E.~N. 1955, \apj, 122, 293

\bibitem[{{Parker}(1975)}]{Par75}
---. 1975, \apj, 198, 205

\bibitem[{{Perdelwitz} {et~al.}(2021){Perdelwitz}, {Mittag}, {Tal-Or},
  {Schmitt}, {Caballero}, {Jeffers}, {Reiners}, {Schweitzer}, {Trifonov},
  {Ribas}, {Quirrenbach}, {Amado}, {Seifert}, {Cifuentes},
  {Cort{\'e}s-Contreras}, {Montes}, {Revilla}, \&
  {Skrzypinski}}]{Perdelwitz2021}
{Perdelwitz}, V., {Mittag}, M., {Tal-Or}, L., {et~al.} 2021, \aap, 652, A116

\bibitem[{{Pizzolato} {et~al.}(2000){Pizzolato}, {Maggio}, {Micela},
  {Sciortino}, {Ventura}, \& {D'Antona}}]{Pizzolato2000}
{Pizzolato}, N., {Maggio}, A., {Micela}, G., {et~al.} 2000, in Astronomical
  Society of the Pacific Conference Series, Vol. 198, Stellar Clusters and
  Associations: Convection, Rotation, and Dynamos, ed. R.~{Pallavicini},
  G.~{Micela}, \& S.~{Sciortino}, 71

\bibitem[{{Rauscher} \& {Marcy}(2006)}]{Rauscher2006}
{Rauscher}, E., \& {Marcy}, G.~W. 2006, \pasp, 118, 617

\bibitem[{{Reiners} {et~al.}(2014){Reiners}, {Sch{\"u}ssler}, \&
  {Passegger}}]{Reiners2014}
{Reiners}, A., {Sch{\"u}ssler}, M., \& {Passegger}, V.~M. 2014, \apj, 794, 144

\bibitem[{Robertson {et~al.}(2014)Robertson, Mahadevan, Endl, \& Roy}]{Rob14}
Robertson, P., Mahadevan, S., Endl, M., \& Roy, A. 2014, Science, 345, 440

\bibitem[{{Rutten}(1984)}]{Rutten1984}
{Rutten}, R.~G.~M. 1984, \aap, 130, 353

\bibitem[{Salvatier {et~al.}(2016)Salvatier, Wiecki, \&
  Fonnesbeck}]{Salvatier2016}
Salvatier, J., Wiecki, T.~V., \& Fonnesbeck, C. 2016, {PeerJ} Computer Science,
  2, e55.
\newblock \url{https://doi.org/10.7717/peerj-cs.55}

\bibitem[{{Schmidt} {et~al.}(2014){Schmidt}, {Prieto}, {Stanek}, {Shappee},
  {Morrell}, {Bardalez Gagliuffi}, {Kochanek}, {Jencson}, {Holoien}, {Basu},
  {Beacom}, {Szczygie{\l}}, {Pojmanski}, {Brimacombe}, {Dubberley}, {Elphick},
  {Foale}, {Hawkins}, {Mullins}, {Rosing}, {Ross}, \& {Walker}}]{Sch14}
{Schmidt}, S.~J., {Prieto}, J.~L., {Stanek}, K.~Z., {et~al.} 2014, \apjl, 781,
  L24

\bibitem[{{Schrijver} \& {Rutten}(1987)}]{Schrijver1987}
{Schrijver}, C.~J., \& {Rutten}, R.~G.~M. 1987, \aap, 177, 143

\bibitem[{{Schr{\"o}der} {et~al.}(2009){Schr{\"o}der}, {Reiners}, \&
  {Schmitt}}]{Schroder2009}
{Schr{\"o}der}, C., {Reiners}, A., \& {Schmitt}, J.~H.~M.~M. 2009, \aap, 493,
  1099

\bibitem[{{Shields} {et~al.}(2016){Shields}, {Ballard}, \& {Johnson}}]{Shi16}
{Shields}, A.~L., {Ballard}, S., \& {Johnson}, J.~A. 2016, \physrep, 663, 1

\bibitem[{{Skumanich}(1972)}]{Skumanich1972}
{Skumanich}, A. 1972, \apj, 171, 565

\bibitem[{{Su{\'a}rez Mascare{\~n}o} {et~al.}(2016){Su{\'a}rez Mascare{\~n}o},
  {Rebolo}, \& {Gonz{\'a}lez Hern{\'a}ndez}}]{SuarezMascareno2016}
{Su{\'a}rez Mascare{\~n}o}, A., {Rebolo}, R., \& {Gonz{\'a}lez Hern{\'a}ndez},
  J.~I. 2016, \aap, 595, A12

\bibitem[{{Su{\'a}rez Mascare{\~n}o} {et~al.}(2015){Su{\'a}rez Mascare{\~n}o},
  {Rebolo}, {Gonz{\'a}lez Hern{\'a}ndez}, \& {Esposito}}]{Sua15}
{Su{\'a}rez Mascare{\~n}o}, A., {Rebolo}, R., {Gonz{\'a}lez Hern{\'a}ndez},
  J.~I., \& {Esposito}, M. 2015, \mnras, 452, 2745

\bibitem[{{Thompson} {et~al.}(1996){Thompson}, {Toomre}, {Anderson}, {Antia},
  {Berthomieu}, {Burtonclay}, {Chitre}, {Christensen-Dalsgaard}, {Corbard}, {De
  Rosa}, {Genovese}, {Gough}, {Haber}, {Harvey}, {Hill}, {Howe}, {Korzennik},
  {Kosovichev}, {Leibacher}, {Pijpers}, {Provost}, {Rhodes}, {Schou}, {Sekii},
  {Stark}, \& {Wilson}}]{Tho96}
{Thompson}, M.~J., {Toomre}, J., {Anderson}, E.~R., {et~al.} 1996, Science,
  272, 1300

\bibitem[{{Tomczyk} {et~al.}(1996){Tomczyk}, {Schou}, \& {Thompson}}]{Tom96}
{Tomczyk}, S., {Schou}, J., \& {Thompson}, M.~J. 1996, in American Astronomical
  Society Meeting Abstracts, Vol. 188, American Astronomical Society Meeting
  Abstracts \#188, 69.03

\bibitem[{{Torres} {et~al.}(2006){Torres}, {Quast}, {da Silva}, {de La Reza},
  {Melo}, \& {Sterzik}}]{Tor06}
{Torres}, C.~A.~O., {Quast}, G.~R., {da Silva}, L., {et~al.} 2006, \aap, 460,
  695

\bibitem[{Vanderburg {et~al.}(2016)Vanderburg, Plavchan, Johnson, Ciardi,
  Swift, \& Kane}]{Van16}
Vanderburg, A., Plavchan, P., Johnson, J.~A., {et~al.} 2016, Monthly Notices of
  the Royal Astronomical Society, 459, 3565

\bibitem[{{Vaughan} {et~al.}(1981){Vaughan}, {Baliunas}, {Middelkoop},
  {Hartmann}, {Mihalas}, {Noyes}, \& {Preston}}]{Vau81}
{Vaughan}, A.~H., {Baliunas}, S.~L., {Middelkoop}, F., {et~al.} 1981, \apj,
  250, 276

\bibitem[{Vaughan \& Preston(1980)}]{vaughan1980}
Vaughan, A.~H., \& Preston, G.~W. 1980, Publications of the Astronomical
  Society of the Pacific, 92, 385

\bibitem[{{Vilhu}(1984)}]{Vilhu1984}
{Vilhu}, O. 1984, \aap, 133, 117

\bibitem[{{West} {et~al.}(2015){West}, {Weisenburger}, {Irwin},
  {Berta-Thompson}, {Charbonneau}, {Dittmann}, \& {Pineda}}]{West2015}
{West}, A.~A., {Weisenburger}, K.~L., {Irwin}, J., {et~al.} 2015, \apj, 812, 3

\bibitem[{{Wilson}(1963)}]{Wil63}
{Wilson}, O.~C. 1963, \apj, 138, 832

\bibitem[{{Wilson}(1968)}]{Wil68}
---. 1968, \apj, 153, 221

\bibitem[{Wright {et~al.}(2011)Wright, Drake, Mamajek, \& Henry}]{Wri11}
Wright, N.~J., Drake, J.~J., Mamajek, E.~E., \& Henry, G.~W. 2011, The
  Astrophysical Journal, 743, 48.
\newblock \url{https://doi.org/10.1088%2F0004-637x%2F743%2F1%2F48}

\bibitem[{{Wright} {et~al.}(2018){Wright}, {Newton}, {Williams}, {Drake}, \&
  {Yadav}}]{Wri18}
{Wright}, N.~J., {Newton}, E.~R., {Williams}, P. K.~G., {Drake}, J.~J., \&
  {Yadav}, R.~K. 2018, \mnras, 479, 2351

\end{thebibliography}

\section{Analysis}\label{sec:Analysis}
Since the early 1960s, the Calcium Fraunhaufer lines have been used as
chromospheric activity tracers \citep{Wil63}. Ca II H\&K lines are observed as
a combination of a broad absorption feature originating in the upper
photosphere along with a narrow emission feature from non-thermal heating of
the upper chromosphere \citep{Catalano1983}. Specifically, the ratio between
emission in the Ca II H\&K lines and flux contributed from the photosphere is
used to define an activity metric known as the S-index \citep{Wil68}.   The
S-index increases with increasing magnetic activity. The S-index is defined as 

\begin{equation}\label{eqn:SIndex}
    S = \alpha \frac{f_{H} + f_{K}}{f_{V} + f_{R}}    
\end{equation}

\noindent where $f_{H}$  and $f_{K}$ are the integrated flux over triangular
passbands with a full width at half maximum of $1.09\text{ \AA}$ centered at
$3968.47\text{ \AA}$ and $3933.66\text{ \AA}$, respectively. The values of
$f_{V}$ and $f_{R}$ are integrated, top hat, broadband regions. They
approximate the continuum (Figure \ref{fig:SindexBandpass}) and are centered at
3901 \AA \ and 4001 \AA \ respectively, with widths of 20 \AA \ each. Finally,
$\alpha$ is a scaling factor with $\alpha = 2.4$.

\begin{figure*}[ht!]
    \centering
    \includegraphics[width=0.9\textwidth]{SIndexBandpass.pdf}
	\caption{Spectrum of 2MASS J06105288-4324178 overplotted with the S index
	bandpasses. (top) V band and Ca II K emission line. (bottom) Ca II H
	emission line and R band. Note that the rectangular and triangular
	regions denote both the wavelength range of the band and the relative
	weight assigned to each wavelength in the band while integrating. }
    \label{fig:SindexBandpass}
\end{figure*}

Following the procedure outlined in \citet{Lov11} we use the mean flux per
wavelength interval, $\tilde{f_{i}}$, as opposed to the integrated flux over
each passband when computing the S-index. This means that for each passband,
$i$, with a blue most wavelength $\lambda_{b,i}$ and a red most wavelength
$\lambda_{r,i}$, $\tilde{f}_{i}$ is the summation of the product of flux ($f$)
and weight ($w_{i})$ over the passband.

\begin{equation}\label{eqn:meanFlux}
    \tilde{f}_{i} = \frac{\sum_{l = \lambda_{b,i}}^{\lambda_{r,i}}f(l)w_{i}(l)}{\lambda_{r,i}-\lambda_{b,i}}    
\end{equation}
\noindent where $w_{i}$ represents the triangular passband for $f_{H}$ \& $f_{K}$ and the tophat for $f_{V}$ \& $f_{R}$.

Additionally, the spectrograph used at Mount Wilson during the development of
the S-index exposed the H \& K lines for eight times longer than the continuum
of the spectra. Therefore, for a modern instrument that exposes the entire
sensor simultaneously, there will be 8 times less flux in the Ca II H\&K
passbands than the continuum passbands than for historical observations. This
additional flux is accounted for by defining a new constant $\alpha_{H}$,
defined as:

\begin{equation}
    \alpha_{H} = 8\alpha\left(\frac{1.09\text{ \AA}}{20\text{ \AA}}\right)
\end{equation}
Therefore, S-indices are calculated here not based on the historical definition
given in Equation \ref{eqn:SIndex}; rather, the slightly modified version:

\begin{equation}\label{eqn:finalSIndex}
    S = \alpha_{H}\frac{\tilde{f}_{H} + \tilde{f}_{K}}{\tilde{f}_{V} + \tilde{f}_{R}}
\end{equation}

The S-index may be used to make meaningful comparisons between stars of similar
spectral class; however, it does not account for variations in photospheric
flux and is therefore inadequate for making comparisons between stars of
different spectral classes. The $R'_{HK}$ index \citep{Middelkoop1982} is a
transformation of the S-index intended to remove the contribution of the
photosphere. 

$R'_{HK}$ introduces a bolometric correction factor, $C_{cf}$, developed by
\citet{Middelkoop1982} and later improved upon by \citet{Rutten1984}.
Calibrations of $C_{cf}$ have focused on FGK-type stars using broad band color
indices, predominately B-V. However, these FGK-type solutions do not extend to
later type stars easily as many mid-late M-dwarfs lack B-V photometry.
Consequently, $C_{cf}$ based on B-V colors were never calibrated for M-dwarfs
as many M-dwarfs lack B and V photometry. \citet{SuarezMascareno2016} provided
the first $C_{cf}$ calibrations for M-dwarfs using the more appropriate color
index of $V-K$. The calibration was later extended by \citet{Def17}, which we
adopt here. 

Generally $R'_{HK}$ is defined as

\begin{equation}\label{eqn:RpHKDef}
    R'_{HK} = K\sigma^{-1}10^{-14}C_{cf}(S-S_{phot})
\end{equation}
where K is a factor to scale surface fluxes of arbitrary units into physical
units; the current best value for K is taken from \citet{Hal07},
$K=1.07\times10^{6}\text{erg cm$^{-2}$ s$^{-1}$}$. $S_{phot}$ is the
photospheric contribution to the S-index; in the spectra this manifests as the
broad absorption feature wherein the narrow Ca II H\&K emission resides.
$\sigma$ is the Stephan-Boltzmann constant. If we define 

\begin{equation}
    R_{phot}\equiv K\sigma^{-1}10^{-14}C_{cf}S_{phot}
\end{equation}
then we may write $R'_{HK}$ as 

\begin{equation}\label{eqn:RpHKFinal}
    R'_{HK} = K\sigma^{-1}10^{-14}C_{cf}S - R_{phot}.
\end{equation}

We use the color calibrated coefficients for $\log_{10}(C_{cf})$ and
$\log_{10}(R_{phot})$ presented in Table 1 of \citet{Def17}.

We estimate the uncertainty of $R'_{HK}$ as the standard deviation of a
distribution of $R'_{HK}$ measurements from 5000 Monte Carlo tests. For each
science target we offset the flux value at each wavelength bin by an amount
sampled from a normal distribution. The standard deviation of this normal
distribution is equal to the estimated error at each wavelength bin. These
errors are calculated at reduction time by the pipeline. The R$'_{HK}$
uncertainty varies drastically with signal-to-noise; targets with
signal-to-noise ratios $\sim 5$ have typical uncertainties of a few percent
whereas targets with signal-to-noise ratios $\sim 100$ have typical
uncertainties of a few tenth of a percent.

\subsection{Rotation and Rossby Number}
The goal of this work is to constrain the rotation activity relation;
therefore, in addition to the measured $R'_{HK}$ value, we also need the
rotation of the star. As mentioned, one of the selection criteria for targets
was that their rotation periods were already measured; however, ultimately
6 of the 53 targets with acceptable S/N did not have well constrained
rotational periods. We therefore only use the remaining 47 targets to fit the
rotation-activity relation. 

% In order to make the most meaningful comparison possible we transform rotation
% period into Rossby Number . This transformation was done using the convective
% overturn timescale, $\tau_{c}$, such that the Rossby Number, $Ro =
% \frac{P_{rot}}{\tau_{c}}$ . To first order $\tau_{c}$ can be approximated as
% $70$ days for fully-convective M-dwarfs \citep{Pizzolato2000}. However,
% \citet{Wri18} Equation (5) presents an empirically calibrated expression for
% $\tau_{c}$. This calibration is derived by fitting the convective overturn
% timescale as a function of color index, in order to minimize the horizontal
% offset between stars of different mass in the rotation-activity relationship.
% The calibration from \citet{Wri18} that we use to find convective overturn
% timescales and subsequently Rossby numbers is:

In order to make the most meaningful comparison possible we transform rotation
period into Rossby Number . This transformation was done using the convective
overturn timescale, $\tau_{c}$, such that the Rossby Number, $Ro =
P_{rot}/\tau_{c}$ . To first order $\tau_{c}$ can be approximated as $70$ days
for fully-convective M-dwarfs \citep{Pizzolato2000}. However, \citet{Wri18}
Equation (5) presents an empirically calibrated expression for $\tau_{c}$. This
calibration is derived by fitting the convective overturn timescale as a
function of color index, in order to minimize the horizontal offset between
stars of different mass in the rotation-activity relationship.  The calibration
from \citet{Wri18} that we use to find convective overturn timescales and
subsequently Rossby numbers is:

\begin{equation}\label{eqn:convectiveOverturn}
    \log_{10}(\tau_{c}) = (0.64\pm0.12)+(0.25\pm0.08)(V-K)
\end{equation}
We adopt symmetric errors for the parameters of Equation
\ref{eqn:convectiveOverturn} equal to the larger of the two anti-symmetric
errors presented in \citet{Wri18} Equation 5. 
